%\documentclass[preprint2]{aastex}
\documentclass[12pt,preprint]{aastex}
%\documentclass[twocolumn,twoside]{article}
\setlength{\textheight}{9.0in}
\setlength{\topmargin}{-0.5in}
%\usepackage{emulateapj5}

\begin{document}

%\begin{frontmatter}
\title{Current Milagro Author List\\
Last Revised \today}

%  
%   To reorder the numbering, e.g. because a name is not used or an extra name is added
%   Do not change any of the \ref{xxx} attached to the names.
%   Reorder the \altaffiltex{N}{\label{xxx} Address} entries, and then renumber them changing
%  only the N in each line
%
\author{
A.~A.~Abdo,\altaffilmark{\ref{msu},\ref{nrl}}
B.~T.~Allen,\altaffilmark{\ref{uci},\ref{cfa}} 
%R.~Atkins,\altaffilmark{\ref{wisc},\ref{atkinscurrent}} 
T.~Aune,\altaffilmark{\ref{ucsc}}
%W.~Benbow,\altaffilmark{\ref{ucsc},\ref{benbowcurrent}} 
D.~Berley,\altaffilmark{\ref{umcp}} 
%E.~Blaufuss,\altaffilmark{\ref{umcp}} 
%J.~Bussons,\altaffilmark{\ref{umcp},\ref{bussycurrent}} 
%S.~Casanova,\altaffilmark{\ref{lanl}} 
C.~Chen,\altaffilmark{\ref{uci},\ref{chicago}}
G.~E.~Christopher,\altaffilmark{\ref{nyu}}
%D.~G.~Coyne,\altaffilmark{\ref{ucsc}} 
%R.~S.~Delay,\altaffilmark{\ref{uci}} 
T.~DeYoung,\altaffilmark{\ref{psu}} 
B.~L.~Dingus,\altaffilmark{\ref{lanl}} 
%D.~E.~Dorfan,\altaffilmark{\ref{ucsc}} 
R.~W.~Ellsworth,\altaffilmark{\ref{georgemason}} 
%A.~Falcone,\altaffilmark{\ref{unh}, 17} 
%L.~Fleysher,\altaffilmark{\ref{nyu}} 
%R.~Fleysher,\altaffilmark{\ref{nyu}} 
%G.~Gisler,\altaffilmark{\ref{lanl}} 
M.~M.~Gonzalez,\altaffilmark{\ref{ida}} 
J.~A.~Goodman,\altaffilmark{\ref{umcp}} 
%T.~J.~Haines,\altaffilmark{\ref{lanl}} 
E.~Hays,\altaffilmark{\ref{gsfc}} 
C.~M.~Hoffman,\altaffilmark{\ref{lanl}}
%B.~Hopper,\altaffilmark{\ref{umcp}}
P.~H.~H\"untemeyer,\altaffilmark{\ref{utah}}
%L.~A.~Kelley,\altaffilmark{\ref{ucsc}} 
B.~E.~Kolterman,\altaffilmark{\ref{nyu}} 
%C.~P.~Lansdell,\altaffilmark{\ref{umcp}}
J.~T.~Linnemann,\altaffilmark{\ref{msu}}
%J.~McCullough,\altaffilmark{\ref{ucsc},18} 
J.~E.~McEnery,\altaffilmark{\ref{gsfc}}
%R.~S.~Miller,\altaffilmark{\ref{unh}}
A.~I.~Mincer,\altaffilmark{\ref{nyu}} 
%M.~F.~Morales,\altaffilmark{\ref{ucsc},\ref{moralescurrent}} 
T.~Morgan,\altaffilmark{\ref{unh}}
P.~Nemethy,\altaffilmark{\ref{nyu}} 
%D.~Noyes,\altaffilmark{\ref{umcp}} 
J.~Pretz,\altaffilmark{\ref{lanl}}
J.~M.~Ryan,\altaffilmark{\ref{unh}} 
%F.~W.~Samuelson,\altaffilmark{\ref{lanl},\ref{samuelsoncurrent}} 
P.~M.~Saz~Parkinson,\altaffilmark{\ref{ucsc}}
%M.~Schneider,\altaffilmark{\ref{ucsc}} 
%B.~Shen,\altaffilmark{16} 
A.~Shoup,\altaffilmark{\ref{osu}} 
G.~Sinnis,\altaffilmark{\ref{lanl}} 
A.~J.~Smith,\altaffilmark{\ref{umcp}} 
%G.~W.~Sullivan,\altaffilmark{\ref{umcp}} 
V.~Vasileiou,\altaffilmark{\ref{umcp},\ref{cresst}} 
G.~P.~Walker,\altaffilmark{\ref{lanl},\ref{nst}} 
%K.~Wang,\altaffilmark{14,19} 
%M.~Wascko,\altaffilmark{14,20} 
D.~A.~Williams\altaffilmark{\ref{ucsc}}
%S.~Westerhoff,\altaffilmark{\ref{ucsc},\ref{westerhoffcurrent}} 
%M.~E.~Wilson,\altaffilmark{\ref{wisc}} 
%X.~W.~Xu\altaffilmark{\ref{lanl}} 
and 
G.~B.~Yodh\altaffilmark{\ref{uci}}} 

\altaffiltext{1}{\label{msu} Department of Physics and Astronomy, Michigan State University, 3245 BioMedical Physical Sciences Building, East Lansing, MI 48824}
\altaffiltext{2}{\label{uci} Department of Physics and Astronomy, University of California, Irvine, CA 92697}
\altaffiltext{3}{\label{ucsc} Santa Cruz Institute for Particle Physics, University of California, 1156 High Street, Santa Cruz, CA 95064}
\altaffiltext{4}{\label{umcp} Department of Physics, University of Maryland, College Park, MD 20742}
\altaffiltext{5}{\label{nyu} Department of Physics, New York University, 4 Washington Place, New York, NY 10003}
\altaffiltext{6}{\label{psu} Department of Physics, Pennsylvania State University, University Park, PA 16802}
\altaffiltext{7}{\label{lanl} Group P-23, Los Alamos National Laboratory, P.O. Box 1663, Los Alamos, NM 87545}
\altaffiltext{8}{\label{georgemason} Department of Physics and Astronomy, George Mason University, 4400 University Drive, Fairfax, VA 22030}
\altaffiltext{9}{\label{ida} Instituto de Astronom\'ia, Universidad Nacional Aut\'onoma de M\'exico,
D.F., M\'exico, 04510}
\altaffiltext{10}{\label{gsfc} NASA Goddard Space Flight Center, Greenbelt, MD 20771}
\altaffiltext{11}{\label{utah} Department of Physics, University of Utah, Salt Lake City, UT 84112}
\altaffiltext{12}{\label{unh} Department of Physics, University of New Hampshire, Morse Hall, Durham, NH 03824} %-3525}
\altaffiltext{13}{\label{osu} Ohio State University, Lima, OH 45804}
\altaffiltext{14}{\label{nrl} Current address: Space Science Division, Naval Research Laboratory, Washington, DC 20375}
\altaffiltext{15}{\label{cfa} Current address: Harvard-Smithsonian Center for Astrophysics, Cambridge, MA 02138}
\altaffiltext{16}{\label{chicago} Current address: University of Chicago, Chicago, IL 60637}
\altaffiltext{17}{\label{cresst} Current address: CRESST NASA/Goddard Space Flight Center, MD 20771 and University of Maryland, Baltimore County, MD 21250}
\altaffiltext{18}{\label{nst} Current address: National Security Technologies, Las Vegas, NV 89102}
%\altaffiltext{12}{\label{samuelsoncurrent} Current address: Office of Science and Engineering Laboratories, Center for Devices and Radiological Health, U.S. Food and Drug Administration}
%\altaffiltext{8}{\label{wisc} Department of Physics, University of Wisconsin, 1150 University Ave, Madison, WI 53706}
%\altaffiltext{4}{\label{atkinscurrent} Current address: Department of Physics, University of Utah, 115 South 1400 East, Salt Lake City, UT 84112}
%\altaffiltext{11}{\label{moralescurrent} Current address: Massachusetts Institute of Technology, Building 37-664H, 77 Massachusetts Avenue, Cambridge, MA 02139}
%\altaffiltext{6}{\label{benbowcurrent} Current address:  Max-Planck-Institut f\"ur Kernphysik, Postfach 103980, D-69029 Heidelberg, Germany}
% \altaffiltext{6}{\label{bussycurrent} Current address: Universit\'e de Montpellier II, Montpellier Cedex 5, F-34095, France}
% \altaffiltext{15}{\label{westerhoffcurrent} Current address: Department of Physics, Columbia University, 538 West 120th Street,  New York, NY 10027}
%\altaffiltext{16}{University of California, Riverside, CA 92521}
%\altaffiltext{17}{Current address: Purdue University, West Lafayette, IN 47907}
%\altaffiltext{18}{Current address: Cabrillo College, Aptos, CA 95003}
%\altaffiltext{19}{Current address: Stanford Linear Accelerator Center, Stanford, CA 94309}
%\altaffiltext{20}{Current address: Louisiana State University, Baton Rouge, LA 70803}
%\altaffiltext{11}{\label{hayscurrent} Current address: High Energy Physics Division, Argonne National Laboratory, Argonne, IL 60439 and Enrico Fermi Institute, University of Chicago, Chicago, IL   60637}

%\maketitle
\section*{Acknowledgments}

We acknowledge Scott Delay and Michael Schneider for their dedicated efforts in the 
construction and maintenance of the Milagro experiment.  This work has been supported by the 
National Science Foundation (under grants 
%-0096256, %UW-Madison
%-0097315, %LANL via UMD 
PHY-0245234, %UCI; previous is PHY-0070933
-0302000, %UMD
-0400424, %Milagro Operations; previous is PHY-0075326
-0504201, %NYU current; previous are PHY-9901496 & PHY-0206656
-0601080, %UCSC; previous is PHY-0070927 & PHY-0245143
and
ATM-0002744) %UNH
the US Department of Energy (Office of High-Energy Physics and 
Office of Nuclear Physics), Los Alamos National Laboratory, the University of
California, and the Institute of Geophysics and Planetary Physics.  



\end{document}




